Title: German ambulatory care physicians’ perspectives on clinical guidelines - a national survey

Authors:

Martin Butzlaff (butzlaff@uni-wh.de)
Daniela Kempkens (daniela.kempkens@uni-wh.de)
Melanie Schnee (melanie.schnee@bertelsmann.de)
Wilfried E Dieterle (wd@psysom.ukl.uni-freiburg.de)
Jan Bocken (jan.boecken@bertelsmann.de)
Monika A Rieger (monika.rieger@uni-wh.de)

Version: 3 Date: 20 July 2006

Author’s response to reviews:

Dear editorial team,
the following final changes were made:
  1. names listed under each affiliation were removed
  2. numbers were removed from section headings
  3. a conclusions section was inserted
  4. contributors section was renamed
  5. ethics information was included in methods section
  6. figure legend colour explanations are given textually
  7. figure titles were removed from the image files